Author's response to reviews

Title: Protocol for a randomised controlled trial of a decision aid for the management of pain in labour and childbirth [ISRCTN52287533]

Authors:

Christine L Roberts (christine.roberts@perinatal.usyd.edu.au)
Camille H Raynes-Greenow (camillerg@perinatal.usyd.edu.au)
Natasha Nassar (nnassar@perinatal.usyd.edu.au)
Lyndal Trevena (lyndalt@health.usyd.edu.au)
Kirsten McCaffery (kirstenm@health.usyd.edu.au)

Version: 2 Date: 29 November 2004

Author's response to reviews:

29 November 2004

The Editorial Board,
BMC Pregnancy and Childbirth
Middlesex House, 34-42 Cleveland Street
London W1T 4LB UK

To the Editorial Board,

Re: Protocol for a randomised controlled trial of a decision aid for the management of pain in labour and childbirth [ISRCTN52287533]

Thank you for accepting the above manuscript for publication in BMC Pregnancy and Childbirth.

We have made the following changes as requested:

Title page - 'Study Protocol' has been removed from the title page.
Abstract - is structured as Background, Methods/Design, Discussion and has been updated on the submission page.
Author list - Camille Raynes-Greenow's middle initial has been added to the manuscript.
Email addresses - are listed below the affiliations on the title page.
References - References are cited in the text using consecutive numbers in square brackets and are placed before the full stop.
Section headings - There is no capitalisation in headings.
Manuscript sections - Manuscript sections should include Abstract; Background; Methods/Design; Competing interests; Authors' contributions; Acknowledgements; References; Figure legend
Figure - Has been uploaded as a separate file.

Please do not hesitate to contact me if you require any further information.

Yours sincerely,

Dr Christine Roberts